UC Davis

UC Davis Previously Published Works

Title

Serum 25-hydroxyvitamin D concentrations and risk for hip fractures.

Permalink https://escholarship.org/uc/item/3hg5155b

Journal Annals of internal medicine, 149(4)

ISSN 0003-4819

Authors

Cauley, Jane A Lacroix, Andrea Z Wu, LieLing <u>et al.</u>

Publication Date 2008-08-01

_

DOI 10.7326/0003-4819-149-4-200808190-00005

Peer reviewed

ARTICLE

Annals of Internal Medicine

Serum 25-Hydroxyvitamin D Concentrations and Risk for Hip Fractures

Jane A. Cauley, DrPH; Andrea Z. LaCroix, PhD; LieLing Wu, MS; Mara Horwitz, MD; Michelle E. Danielson, PhD; Doug C. Bauer, MD; Jennifer S. Lee, MD; Rebecca D. Jackson, MD; John A. Robbins, MD; Chunyuan Wu, MS; Frank Z. Stanczyk, PhD; Meryl S. LeBoff, MD; Jean Wactawski-Wende, PhD; Gloria Sarto, MD; Judith Ockene, PhD; and Steven R. Cummings, MD

Background: The relationship between serum 25-hydroxyvitamin D [25(OH) vitamin D] concentration and hip fractures is unclear.

Objective: To see whether low serum 25(OH) vitamin D concentrations are associated with hip fractures in community-dwelling women.

Design: Nested case-control study.

Setting: 40 clinical centers in the United States.

Participants: 400 case-patients with incident hip fracture and 400 control participants matched on the basis of age, race or ethnicity, and date of blood draw. Both groups were selected from 39 795 postmenopausal women who were not using estrogens or other bone-active therapies and who had not had a previous hip fracture.

Measurements: Serum 25(OH) vitamin D was measured and patients were followed for a median of 7.1 years (range, 0.7 to 9.3 years) to assess fractures.

Results: Mean serum 25(OH) vitamin D concentrations were lower in case-patients than in control participants (55.95 nmol/L [SD, 20.28] vs. 59.60 nmol/L [SD, 18.05]; P = 0.007), and lower serum 25(OH) vitamin D concentrations increased hip fracture risk (adjusted odds ratio for each 25-nmol/L decrease, 1.33 [95% CI, 1.06 to 1.68]). Women with the lowest 25(OH) vitamin D concentrations (\leq 47.5 nmol/L) had a higher fracture risk than did those with the highest concentrations (\geq 70.7 nmol/L) (adjusted odds ratio, 1.71 [CI, 1.05 to 2.79]), and the risk increased statistically significantly across quartiles of serum 25(OH) vitamin D concentration (P for trend = 0.016). This association was independent of number of falls, physical function, frailty, renal function, and sex-steroid hormone levels and seemed to be partially mediated by bone resorption.

Limitations: Few case-patients were nonwhite women. Bone mineral density and parathyroid hormone levels were not accounted for in the analysis.

Conclusion: Low serum 25(OH) vitamin D concentrations are associated with a higher risk for hip fracture.

www.annals.org

Ann Intern Med. 2008;149:242-250. For author affiliations, see end of text.

Vitamin D deficiency is common in older adults, especially during the winter (1) and in homebound populations (2), general medical inpatients (3), and communitydwelling women admitted to the hospital with acute hip fracture (4). A recently published evidence-based report on vitamin D and bone health (5) found the level of evidence for an association between serum 25-hydroxyvitamin D [25(OH) vitamin D] concentrations and fracture risk to be inconsistent (5). Since publication of that review, 1 prospective study (6) reported no relationship between serum 25(OH) vitamin D concentrations and fractures, whereas another (7) reported a significantly lower risk for hip fracture with 25(OH) vitamin D concentrations greater than 60 nmol/L.

Vitamin D concentration could be associated with fractures in several ways. It could influence muscle strength and balance, both of which contribute to falls and disability (8–10). The association between 25(OH) vitamin D concentrations and fracture may also be influenced by renal function, because renal insufficiency has been linked to fracture (11) and to vitamin D deficiency (12). Several interactions between vitamin D and estrogen receptors have been described (13); hormone therapy has been shown to reverse abnormalities in vitamin D metabolism (14), and low vitamin D concentrations have also been linked to higher bone turnover (15, 16). Thus, sex-steroid hormones and bone turnover could contribute to the association between 25(OH) vitamin D concentration and fractures.

242 © 2008 American College of Physicians

We conducted a nested case–control study within the WHI-OS (Women's Health Initiative Observational Study) among 400 case-patients with adjudicated incident hip fracture and 400 control participants. We tested whether low serum 25(OH) vitamin D concentrations are associated with a higher risk for hip fractures in community-dwelling women and whether this relationship may be mediated by poor physical functioning, frailty, falls, sex-steroid hormones, renal function, or bone turnover.

METHODS

Study Population

Our study population came from the WHI-OS, a prospective cohort study that enrolled 93 676 women between 1994 and 1998 at 40 U.S. clinical centers (age range, 50 to 79 years). Study methods are described in detail elsewhere (17). In brief, women were eligible if they were postmeno-

See also:

Print

Editors' Notes 24	3
Summary for Patients I-42	2

Web-Only

Appendix Conversion of graphics into slides pausal, were unlikely to move or die within 3 years, were not enrolled in the WHI clinical trials, and were not currently participating in any other clinical trial. The human subjects review committees from each participating institution approved the study.

Follow-up and Outcome Ascertainment

We sent women questionnaires annually to report any hospitalization and other outcomes, including fractures. As of August 2004, median follow-up duration was 7.1 years (range, 0.7 to 9.3 years). At that time, 3.7% of participants had withdrawn or were lost to follow-up and 5.3% had died. We reviewed medical records to verify cases of hip fracture, and blinded central adjudicators confirmed the cases (18). We excluded patients with pathologic hip fractures.

Nested Case-Control Study Design

The present study is a case-control study nested within the prospective design of WHI-OS. We excluded women who had a history of hip fracture; were receiving hormone therapy up to 1 year before enrollment; or were currently receiving androgens, selective estrogen receptor modulators, antiestrogens, or other osteoporosis treatments (bisphosphonates, calcitonin, or parathyroid hormone). We also excluded women with insufficient serum stored or of unknown ethnicity, leaving 39 793 eligible participants. Of these, 404 women had a hip fracture. We randomly selected 400 of these women to form the incident hip fracture group. For each case-patient, we selected 1 control participant who was within 1 year of the case-patient's age at screening, was of matching race or ethnicity, and had their blood drawn within 120 days of the case-patient's blood draw date; 99% of case-patients and control participants were matched within 30 days.

Baseline Clinical Variables

We divided clinical centers into 3 geographic regions on the basis of latitude: northern (>40 °N), middle (35 to 40 °N), and southern (<35 °N). We ascertained all covariates at baseline. Clinic interviewers recorded current use of prescription medications by direct inspection of medicine containers. We entered prescription names into the WHI database and assigned drug codes by using Medispan software (First DataBank, San Bruno, California). Average amounts of elemental calcium and vitamin D preparations were entered directly from supplement containers. Dietary intakes of calcium and vitamin D were assessed by using a semiquantitative food-frequency questionnaire (19). Total calcium and vitamin D intake was defined as the sum of diet and supplements.

We used questionnaires to ascertain date of birth, race or ethnicity, age at menopause, history of any fracture after age 55 years, smoking, parental history of hip fracture, self-rated health status, and alcohol consumption. We classfied physical activity on the basis of frequency and duration of walking and mild, moderate, and strenuous activities in the previous week. We calculated kilocalories

Context

Vitamin D supplementation may help prevent fractures, but the relationship between blood vitamin D concentrations and fracture risk is unclear.

Contribution

These authors observed an increased risk for hip fracture among women with lower serum 25-hydroxyvitamin D [25(OH) vitamin D] concentrations that was independent of measures of frailty, body mass index, physical function, and falls.

Caution

The authors did not measure bone mineral density (BMD), so they could not determine whether 25(OH) vitamin D concentrations give different information about fracture risk than that offered by BMD.

Implication

Low serum 25(OH) vitamin D concentrations seem to be associated with a higher hip fracture risk.

—The Editors

of energy expended in 1 week as the metabolic equivalent (kcal hours/week per kg) (20). We measured physical function by using the RAND Short Form-36 physical function scale, which comprises 10 items measuring whether health now limits physical function in moderate or vigorous activity (2 items); strength to lift, carry, stoop, bend, or stair climb (4 items); ability to walk various distances without difficulty (3 items); and self-care (1 item) (21). The scale is scored from 0 to 100, with higher scores indicating better physical function. We compared women with a score greater than 90 versus those with a score less than or equal to 90, a cutoff value corresponding to the median score. We computed a frailty score, which included self-reported muscle weakness and impaired walking speed (RAND Short Form-36 physical function scale score <75), exhaustion (RAND Short Form-36 vitality scale score <55), low physical activity (lowest quartile of physical activity), and unintended weight loss between baseline and 3 years of follow-up (22). A woman was considered frail if she reported 3 or more of these indicators. Weight was measured on a balance-beam scale with the participant dressed in indoor clothing without shoes. Height was measured by using a wall-mounted stadiometer. Body mass index was calculated as weight (in kg) divided by height (in m²).

Laboratory Procedures

Laboratory personnel blinded to case–control status obtained a 12-hour fasting blood sample at the baseline visit, which was processed and stored at -80 °C according to strict quality control procedures (23). Serum 25(OH) vitamin D concentrations and sex-steroid hormone levels were measured at the Reproductive Endocrine Research

www.annals.org

Laboratory at the University of Southern California. 25-Hydroxyvitamin vitamin D was measured by using a radioimmunoassay with DiaSorin reagents (DiaSorin, Stillwater, Minnesota). The sensitivity of the assay was 3.75 nmol/L. The interassay coefficients of variation were 11.7%, 10.5%, 8.6%, and 12.5% at 14.0, 56.8, 82.5, and 122.5 nmol/L, respectively.

Estradiol and testosterone concentrations were quantified by using sensitive and specific radioimmunoassays after organic solvent extraction and celite column partition chromatography (24-27). The intra- and interassay coefficients of variation were 7.9% and 8% to 12%, respectively, for estradiol and 6% and 10% to 12%, respectively, for testosterone. We calculated bioavailable hormone concentrations by using mass action equations (28-30). We measured sex hormone-binding globulin by using a solidphase, 2-site chemiluminescent immunoassay. The intraand interassay coefficients of variation were 4.1% to 7.7% and 5.8% to 13%, respectively. Serum cystatin C, a marker of renal function that is independent of age and weight, was measured at Medical Research Laboratories International, Highland Heights, Kentucky, by using the Dade Behring BN-II nephelometer and Dade Behring reagents (Dade Behring, Ramsey, Minnesota) in a particle-enhanced immunonepholometric assay. Serum C-terminal telopeptide of type I collagen and aminoterminal procollagen extensions propeptide were measured by immunoassay (Synarc, Lyon, France).

Statistical Analysis

We used chi-square and t tests to compare baseline characteristics between case-patients with hip fracture and matched control participants. We assigned 25(OH) vitamin D concentrations to quartile categories defined on the basis of the distribution in the control participants. To further assess confounding, we compared baseline characteristics across quartiles of 25(OH) vitamin D concentrations in case-patients and control participants combined. We calculated the P values for trend by using logistic regression and coding the variable of interest as a continuous variable. We assessed the association between serum 25(OH) vitamin D concentrations and incident hip fracture in conditional logistic regression models that retained the matched case-control design. We first examined the unadjusted associations and then adjusted for age, body mass index, parental history of hip fracture, previous fractures, smoking, alcohol use, total calcium intake, oral corticosteroid use, and geographic location. We had information on 25(OH) vitamin D concentrations in 799 individuals.

We calculated odds ratios and 95% CIs from the conditional logistic regression models for every 2.5-nmol/L and 25-nmol/L decrease in 25(OH) vitamin D concentrations and across quartiles. We defined the highest quartile of 25(OH) vitamin D concentration as the reference group. We separately examined the association in women younger than age 70 years and women 70 years of age or older, as well as the interaction of age and 25(OH) vitamin D concentrations. We used nonparametric smoothing techniques to test whether the relationship was linear or a threshold one, at a prespecified threshold of 50 nmol/L.

To investigate mechanisms by which 25(OH) vitamin D concentration might be associated with hip fracture, we added the following variables one at a time to the base model to determine their effect on the association between serum 25(OH) vitamin D concentration and hip fracture: markers for deteriorating health status (poor physical function, frailty score [22], number of falls, sex-steroid hormones, renal function (cystatin C), and bone turnover (serum C-terminal telopeptide of type I collagen and aminoterminal procollagen extensions propeptide). We then adjusted for all the variables simultaneously (except for frailty, which we hypothesized would be correlated with measures of physical function because both rely on the RAND Short Form-36 physical function scale). We hypothesized that the association between 25(OH) vitamin D concentration and hip fracture would be reduced after adjustment for these factors if they are in the causal pathway. The Hosmer-Lemeshow goodness-of-fit test was used to evaluate the models (31). All models indicated excellent fit for the data, and the area under the curve for our summary multivariate model was 0.69.

Our study had 90% power (with a 2-sided α level of 0.05) to detect a 0.14-SD difference in 25(OH) vitamin D concentrations and a relative risk for fracture of 1.6 between case-patients and control participants.

Role of the Funding Source

The WHI program is funded by the National Heart, Lung and Blood Institute, National Institutes of Health, U.S. Department of Health and Human Services. The sponsor played a role in the design and analysis of the WHI.

RESULTS

Participant Characteristics

The average age of the case-patients was 71 years (SD, 6.15); one third was older than 70 years of age and 95% were white (Table 1). Compared with control participants, case-patients had a lower body mass index; performed less physical activity; were more likely to report oral corticosteroid use, fair or poor health status, poor physical function, and smoking; and were more likely to be considered frail. Mean 25(OH) vitamin D concentrations were lower in case-patients than in control participants (mean, 55.95 nmol/L [SD, 20.28] vs. 59.60 nmol/L [SD, 18.05]; P = 0.007). Case-patients had a lower calcium intake and lower sex-steroid hormone levels but higher cystatin C and bone resorption marker levels than did control participants. Hormone therapy use, alcohol intake, use of vitamin D supplements or dietary vitamin D, personal or family history of fracture, geographic location, and bone formation

Table 1. Baseline Characteristics

Characteristic	Control Participants	Case-Patients	P Value	25(OH) Vitamin D Level				
				First QuartileSecond QuartileThird QuartileFourth Quartile(9.2-47.5(47.6-60.1(60.2-70.6(70.7-121.5nmol/L)nmol/L)nmol/L)nmol/L)				
Age >70 y, n (%)	132 (33.0)	132 (33.0)	1.00	84 (34.43)	57 (29.23)	54 (32.34)	68 (35.23)	0.58
White, n (%)	380 (95.0)	380 (95.0)	1.00	224 (91.80)	184 (94.36)	163 (97.60)	188 (97.41)	0.01
Body mass index, n (%)	27.41 (5.13)	26.04 (5.15)	< 0.001	27.70 (5.72)	27.04 (4.76)	26.14 (4.87)	25.75 (25.75)	<0.00
<25 kg/m ²	144 (36.09)	193 (48.61)	0.001	91 (37.45)	70 (36.27)	77 (46.11)	98 (51.04)	0.00
25 to <30 kg/m ²	150 (37.59)	127 (31.99)		80 (32.92)	72 (37.31)	63 (37.72)	62 (32.29)	
\geq 30 kg/m ²	105 (26.32)	77 (19.40)		72 (29.63)	51 (26.42)	27 (16.17)	32 (16.67)	
Previously received hormone therapy, <i>n</i> (%)	98 (24.50)	95 (23.75)	0.80	57 (23.36)	44 (22.56)	48 (28.74)	43 (22.28)	0.45
Oral corticosteroid use, n (%)	3 (0.75)	14 (3.50)	0.007	7 (2.87)	3 (1.54)	1 (0.60)	6 (3.11)	0.29
RAND SF-36 physical function scale score >90, n (%)	117 (29.77)	86 (21.66)	0.009	47 (19.50)	56 (29.02)	38 (23.17)	61 (31.94)	0.01
Mean total physical activity (SD), MET-h/wk	13.86 (15.32)	10.71 (12.67)	0.002	10.31 (15.09)	10.69 (10.37)	12.88 (12.08)	15.92 (17.0)	<0.00
Frailty, <i>n</i> (%)*	66 (16.50)	89 (22.25)	0.040	65 (26.64)	29 (14.87)	31 (18.56)	30 (15.54)	0.00
Fair/poor self-reported health status, n (%)	42 (10.69)	61 (15.37)	0.051	44 (18.26)	24 (12.44)	17 (10.30)	18 (9.47)	0.02
Smoking, <i>n (%)</i>								
Past smoker Current smoker	171 (43.18) 10 (2.53)	144 (36.55) 36 (9.14)	<0.001	92 (37.86) 22 (9.05)	70 (36.65) 10 (5.24)	72 (43.90) 5 (3.05)	81 (42.41) 8 (4.19)	0.12
Mean alcohol intake (SD), <i>drinks/wk</i>	3.40 (1.67)	3.37 (1.60)	0.839	3.29 (1.62)	3.39 (1.67)	3.35 (1.63)	3.51 (1.60)	0.58
Mean 25(OH) vitamin D level (SD), nmol/L	59.60 (18.05)	55.95 (20.28)	0.007	35.83 (8.33)	53.95 (3.73)	65.13 (3.05)	83.00 (10.80)	<0.00
Quartile median	-	-	-	37.1	54.0	65.1	80.4	-
Received vitamin D supplement, n (%)	189 (47.25)	195 (48.75)	0.67	63 (25.82)	96 (49.23)	102 (61.08)	123 (63.73)	<0.00
Mean total vitamin D intake (SD), <i>IU/d</i>	373 (275)	383 (396)	0.707	262 (332)	361 (263)	439 (403)	480 (307)	<0.00
Mean total calcium intake (SD), mg/d	1167.01 (683.85)	1072.47 (694.23)	0.053	906.38 (567.86)	1087.03 (702.50)	1215.80 (731.58)	1341.99 (703.77)	<0.00
History of fracture at or after age 55 y, n (%)	82 (20.50)	96 (24.00)	0.23	49 (20.08)	35 (17.95)	39 (23.35)	55 (28.50)	0.00
Parents had hip fracture, <i>n</i> (%) Geographic region,	64 (16.00)	80 (20.00)	0.141	48 (19.67)	30 (15.38)	28 (16.77)	37 (19.17	0.63
n (%)								
Southern (<35 °N)	108 (27.00)	106 (26.50)	0.24	56 (22.95)	48 (24.62)	48 (28.74)	62 (32.12)	0.0
Middle (35-40 °N)	95 (23.75)	115 (28.75)		59 (24.18)	64 (32.82)	32 (19.16)	54 (27.98)	
Northern (>40 °N) Wean bioavailable estradiol level	197 (49.25) 7.54 (4.54)	179 (44.75) 6.57 (4.34)	0.002	129 (52.87) 7.79 (4.82)	83 (42.56) 7.01 (4.68)	87 (52.10) 6.50 (4.0)	77 (39.90) 6.64 (4.05)	0.0
(SD), pg/mL Wean bioavailable testosterone level (SD), pg/mL	12.59 (6.98)	10.90 (6.26)	<0.001	12.43 (7.21)	11.27 (5.68)	11.43 (6.50)	11.66 (7.04)	0.2
Mean cystatin C level (SD), mg/L	1.06 (0.24)	1.10 (0.30)	0.023	1.12 (0.37)	1.06 (0.21)	1.07 (0.23)	1.04 (0.20)	0.0

Continued on following page

www.annals.org

Table 1—Continued								
Characteristic	Control Participants	Case-Patients	P Value	25(OH) Vitamin D Level				
	Faiticipants			First Quartile (9.2–47.5 nmol/L)	Second Quartile (47.6–60.1 nmol/L)	Third Quartile (60.2–70.6 nmol/L)	Fourth Quartile (70.7–121.5 nmol/L)	
Mean PINP level (SD), ng/mL	49.64 (23.71)	51.00 (23.03)	0.42	49.04 (21.44)	51.65 (21.41)	51.68 (51.68)	49.34 (27.89)	0.53
Mean CTx level (SD), ng/mL	0.41 (0.19)	0.45 (0.21)	0.016	0.45 (0.22)	0.44 (0.19)	0.43 (0.21)	0.39 (0.16)	0.022

25(OH) vitamin D = 25-hydroxyvitamin D; CTx = serum *C*-terminal telopeptide of type I collagen; MET = metabolic equivalent; PINP = amino proterminal procollagen extension propeptide; SF-36 = Short Form-36.

* Frailty was defined as ≥3 of the following: muscle weakness, slow walking speed, exhaustion, low physical activity, and unintentional weight loss (22).

did not differ between case-patients and control participants.

Comparisons across Quartiles of 25(OH) Vitamin D

The percentage of nonwhite, obese, and frail women decreased with increasing 25(OH) vitamin D concentrations, as did the percentage of those with fair or poor health status (**Table 1**). Physical function and physical activity increased with increasing concentration. Use of vitamin D supplements, vitamin D intake, and calcium intake also increased across quartiles of 25(OH) vitamin D concentration. A lower percentage of participants from the northern region were in the highest vitamin D quartile. Sex-steroid, cystatin C, and bone resorption marker levels all decreased with increasing vitamin D concentrations.

25(OH) Vitamin D Concentration and Hip Fracture

Serum 25 (OH) vitamin D concentrations were associated with hip fracture (unadjusted odds ratio for incident hip fracture per 25-nmol/L decrease in concentration, 1.30 [95% CI, 1.07 to 1.58]) (Table 2). Further multivariate adjustment had little effect on the risk estimate. The increased risk for hip fracture was primarily confined to women with the lowest 25(OH) vitamin D concentrations (multivariate-adjusted odds ratio of hip fracture for women in quartile 1 compared with women in quartile 4, 1.71 [CI, 1.05 to 2.79]) (Table 3).

A model testing the specific 25(OH) vitamin D threshold of less than 50 nmol/L did not differ significantly from the linear model (P = 0.78), suggesting a continuous linear relationship between serum 25(OH) vitamin D concentration and hip fracture, at least within the ranges of 25(OH) vitamin D concentrations in our study. The relationship between 25(OH) vitamin D concentration and hip fracture did not differ by age (P for interaction = 0.62).

Potential Mediators

The average number of falls during follow-up did not differ between case-patients (mean, 2.36 [SD, 2.75]) and control participants (mean, 2.82 [SD, 3.5]) (P = 0.09) or across 25(OH) vitamin D concentration quartiles. Adjustment for frailty, physical functioning, and falls resulted in similar small attenuations in the association between 25(OH) vitamin D concentration and hip fractures (**Table** 3). Adjustment for sex-steroid hormones also attenuated the risk estimate, but less so than adjustment for frailty, physical functioning, and falls. Adjustment for serum *C*terminal telopeptide of type I collagen individually and for

Table 2. Odds Ratios of Risk for Hip Fracture*						
25-Hydroxyvitamin D Level	Unadjusted Odds Ratio (95% CI)	Adjusted Odds Ratio (95% CI)†				
Per 2.5-nmol/L decrease‡	1.03 (1.01–1.05)	1.03 (1.01–1.05)				
Per 25-nmol/L decrease	1.30 (1.07–1.58)	1.33 (1.06–1.68)				
Quartile (according to control group)						
First (9.2–47.5 nmol/L)	1.73 (1.13–2.66)	1.71 (1.05–2.79)				
Second (47.6–60.1 nmol/L)	1.08 (0.72–1.63)	1.09 (0.70–1.71)				
Third (60.2–70.6 nmol/L)	0.78 (0.50-1.20)	0.82 (0.51–1.31)				
Fourth (70.7–121.5 nmol/L)	1.00 (reference)	1.00 (reference)				

* We selected and matched case-patients with control participants according to age, race or ethnicity, and blood draw date. Eighteen case-control pairs were missing from our multivariate models because of missing values. We excluded these pairs from our unadjusted models, even when values were not missing, to provide similar risk estimates from the same analytic samples.

* Multivariate adjustment includes age, body mass index, parental history of hip fracture, history of fracture, smoking, alcohol use, total calcium intake, oral corticosteroid use, and geographic region.

‡ P value for linear trend = 0.009 for unadjusted and 0.015 for multivariate-adjusted models; models based on a 25-hydroxyvitamin D threshold level of 50 nmol/L fit no better than the linear models.

Analysis	Odds Ratio (95% CI)						
	First Quartile (9.2–47.5 nmol/L) (n = 244)	Second Quartile (47.6–60.1 nmol/L) (n = 195)	Third Quartile (60.2–70.6 nmol/L) (n = 167)	Fourth Quartile (70.7–121.5 nmol/L) (<i>n</i> = 193)	for Linea Trend		
Base analysis*	1.71 (1.05–2.79)	1.09 (0.70–1.71)	0.82 (0.51–1.31)	1.00 (reference)	0.016		
P value	0.032	0.76	0.45	-	-		
Adjusted analysis							
Frailty scoret	1.65 (1.01-2.69)	1.08 (0.68–1.70)	0.81 (0.50-1.30)	1.00 (reference)	0.022		
P value	0.047	0.74	0.38	-	-		
RAND SF-36 physical functioning scale score >90	1.68 (1.02–2.76)	1.15 (0.73–1.83)	0.81 (0.50-1.31)	1.00 (reference)	0.016		
P value	0.041	0.55	0.39	-	-		
Total number of falls during follow-up	1.66 (1.00–2.77)	1.14 (0.70–1.85)	0.86 (0.52-1.42)	1.00 (reference)	0.018		
P value	0.039	0.59	0.55	-	-		
Bioavailable estradiol	1.75 (1.06–2.87)	1.01 (0.64–1.59)	0.75 (0.46-1.23)	1.00 (reference)	0.022		
P value	0.028	0.98	0.25	-	-		
Bioavailable testosterone	1.72 (1.05–2.83)	1.05 (0.67–1.66)	0.82 (0.50-1.34)	1.00 (reference)	0.021		
P value	0.031	0.83	0.42	-	-		
Cystatin C	1.68 (1.02–2.76)	1.18 (0.70–1.76)	0.81 (0.50-1.31)	1.00 (reference)	0.019		
P value	0.042	0.66	0.39	-	-		
PINP	1.69 (1.04–2.77)	1.05 (0.67–1.67)	0.84 (0.52–1.35)	1.00 (reference)	0.022		
P value	0.036	0.82	0.47	-	-		
CTx	1.58 (0.97–2.60)	1.03 (0.66–1.64)	0.83 (0.51–1.34)	1.00 (reference)	0.040		
P value	0.069	0.89	0.44	-	-		
Summary multivariate model‡	1.72 (0.98–3.02)	1.21 (0.71–2.05)	0.84 (0.48–1.46)	1.00 (reference)	0.029		
P value	0.060	0.48	0.53	_	-		

Table 3. Adjusted Odds Ratios of Risk for Hip Fracture, by Vitamin D Quartile

CTx = serum C-terminal telopeptide of type I collagen; PINP = amino proterminal procollagen extension propeptide; SF-36 = Short Form-36.

* Matched on age, race or ethnicity, and blood draw date and controlled for age, body mass index, parental history of hip fracture, history of fracture, smoking, alcohol use, total calcium intake (from diet, supplements, and medications, per 100-mg increase), oral corticosteroid use, and geographic region.

+ Frailty was defined as ≥3 of the following: muscle weakness, slow walking speed, exhaustion, low physical activity, and unintentional weight loss (23).

* Matched on age, race or ethnicity, and blood draw date and controlled for age; body mass index; parental history of hip fracture; history of fracture; smoking; alcohol use; total calcium intake (from diet, supplement and medications, per 100-mg increase); oral corticosteroid use; geographic region; frailty; total number of falls; and levels of bioavailable estradiol, bioavailable testosterone, cystatin C, PINP, and CTx.

all variables simultaneously resulted in the greatest attenuations in odds ratios and seemed to eliminate the association between 25(OH) vitamin D concentration and hip fracture in women in the lowest 25(OH) vitamin D concentration quartile. However, the overall trend of the relationship between 25(OH) vitamin D concentration and hip fracture across quartiles remained statistically significant (fully adjusted odds of fracture for women with the lowest concentrations [quartile 1], 1.72 [CI, 0.98 to 3.02]; P = 0.060; P for trend = 0.029).

DISCUSSION

In our prospective, nested case–control study, we found that women with the lowest 25(OH) vitamin D concentrations (\leq 47.6 nmol/L) at study entry had a significantly greater increased risk for subsequent hip fracture during the next 7 years than did women with the highest concentrations (\geq 70.7 nmol/L). The association between 25(OH) vitamin D concentration and hip fracture was linear and did not seem to differ by age.

By using an English-language MEDLINE search through January 2008, we identified 5 prospective studies of serum

www.annals.org

vitamin D concentrations and fracture. Our results are consistent with a recent report from the NHANES III (Third National Health and Nutrition Examination Survey) (7), in which the relative risk for hip fracture among participants with 25(OH) vitamin D concentrations greater than 60 nmol/L compared with those who had lower concentrations was 0.64 (CI, 0.46 to 0.89). Similarly, in a cohort study of Swedish women, those with serum 25(OH) vitamin D concentrations less than 52.5 nmol/L had a 2-fold increased risk for fracture (32). Two previous cohort studies (33, 34) failed to find a significant association between 25(OH) vitamin D concentration and fracture; however, these were limited by small sample size and high lost-tofollow-up rates (33) or use of an older assay (34). A nested case-control study of 730 case-patients and 1445 control participants with incident fracture found no evidence of an association between 25(OH) vitamin D concentration and fracture (6). However, the investigators studied a heterogeneous group of fractures, including only 22 hip fractures, and studied a younger population (mean age, 53 years). Vitamin D concentration may be more strongly linked to frailty-related fractures, such as hip fractures, which tend to occur in much older women.

The mechanism for the association is unclear. *C*-terminal telopeptide of type I collagen, a marker of bone resorption, tended to be higher among case-patients with the lowest 25(OH) vitamin D concentrations, an association that may be driven by higher parathyroid hormone levels in this group (2). Adjustment for *C*-terminal telopeptide of type I collagen resulted in the largest attenuation of risk in women with the lowest 25(OH) vitamin D concentrations (unadjusted odds ratio, 1.71 [CI, 1.05 to 2.79]; *C*-terminal telopeptide of type I collagen–adjusted odds ratio, 1.58 [CI, 0.97 to 2.60]), suggesting that high bone resorption is an important mechanism for the association.

The increased fracture risk could also be related to impaired muscle strength and balance and poor physical function, all of which could lead to an increased risk for falls (8–10). Case-patients with hip fractures had lower physical function scores and were more likely to be frail, whereas higher 25(OH) vitamin D concentrations were associated with a lower likelihood of being frail and with higher physical function. Adjustment for physical function, frailty, and falls attenuated slightly the apparent association between 25(OH) vitamin D concentrations and hip fracture; however, the association remained even after we adjusted for these factors. The association also remained after we adjusted for sex-steroid hormone level and measures of renal function.

Finally, we adjusted for geographic location in all our analyses, given the association between vitamin D deficiency and northern latitude (35). Women from northern regions (>40 °N) had lower 25(OH) vitamin D concentrations; however, the relationship between 25(OH) vitamin D concentrations and hip fracture was independent of geographic location. Also, obesity was statistically significantly less prevalent in the participants with the highest 25(OH) vitamin D concentrations, consistent with the lower bioavailability of vitamin D reported in obese patients (36) and with reports of an inverse association between 25(OH) vitamin D concentrations and adiposity (37, 38). Lower 25(OH) vitamin D concentrations in obese patients may reflect lower physical activity (less sunlight exposure) and deposition in body fat compartments (36). However, the association between 25(OH) vitamin D concentration and hip fracture was independent of obesity.

The optimal serum 25(OH) vitamin D concentration needed to maintain bone health has not been established. Optimal concentration has been defined as that at which serum parathyroid hormone levels plateau in the normal range (39); however, this definition has led to a wide range of optimal 25(OH) vitamin D concentration thresholds (20 to 115 nmol/L). More recently (40), the optimal threshold concentration of 25(OH) vitamin D, based on bone mineral density levels, was found to be at least 78 nmol/L, with a target of 92 to 105 nmol/L. Randomized trials of vitamin D supplementation (with or without calcium) that brought mean serum 25(OH) vitamin D concentrations up to 75 to 102.5 nmol/L found significantly lower fracture rates (41). Trials in which the mean serum 25(OH) vitamin D concentration did not reach this threshold showed no overall effect on fractures (42, 43). However, in the largest study to date (43), a 29% (CI, 3% to 48%) reduction in the hazard of hip fracture was observed among adherent women, even though they did not achieve this threshold 25(OH) vitamin D concentration.

The WHI Calcium–Vitamin D Trial (43) found no relationship between serum 25(OH) vitamin D concentration and fracture. However, the characteristics of women in that trial differed from those of the women in our study. More than 50% of the women in the WHI Calcium– Vitamin D Trial were receiving hormone therapy, whereas we excluded women receiving any bone-active agents from our analysis. In addition, the WHI trial used different cutpoints, and the results of the trial analyses were unadjusted.

Our study has several limitations. Although we matched each case-patient to a control participant whose blood draw date was within 120 days of that of the casepatient, seasonal variability in 25(OH) vitamin D concentrations could have confounded our results. Nonwhite women are more likely to be vitamin D-deficient (44); however, only 20 hip fractures occurred in nonwhite women. We could not test whether the association between low 25(OH) vitamin D concentration and hip fracture was mediated by bone mineral density because only 3 WHI clinics measured bone mineral density; however, the NHANES III (7) found this association to be independent of bone density. We measured total 25(OH) vitamin D concentrations and did not distinguish 25(OH) vitamin D_2 from 25(OH) vitamin D_3 . However, reporting vitamin D₂ and vitamin D₃ concentrations separately has been shown to cause some clinical confusion (45). We did not measure parathyroid hormone, which could contribute to the relationship between 25(OH) vitamin D concentrations and hip fracture. The study included few women with 25(OH) vitamin D concentrations greater than 75 nmol/L, so we could not test whether even higher concentrations offer greater protection against hip fracture risk. Finally, we used an observational study design and adjusted for many factors that could confound the association; however, unmeasured factors may have caused residual confounding.

Despite these limitations, we conclude that low serum 25(OH) vitamin D concentrations are associated with an increased risk for hip fracture in community-dwelling women. The mechanism of association is unclear; however, our findings suggest that low serum 25(OH) vitamin D concentrations might help identify women at high risk for hip fracture.

From University of Pittsburgh, Pittsburgh, Pennsylvania; University of Washington, Seattle, Washington; University of California, San Francisco, and San Francisco Coordinating Center, California Pacific Medical Center Research Institute, San Francisco, California; University of Cal-

ifornia, Davis, Davis, California; Ohio State University, Columbus, Ohio; University of Southern California, Los Angeles, California; Brigham and Women's Hospital/Harvard Medical School, Boston, Massachusetts; University at Buffalo, Buffalo, New York; University of Wisconsin-Madison, Madison, Wisconsin; and University of Massachusetts Amherst, Amherst, Massachusetts.

Acknowledgment: The authors thank the WHI investigators. For a list of WHI investigators, see the Appendix, available at www.annals.org.

Grant Support: By the National Heart, Lung and Blood Institute, National Institutes of Health, U.S. Department of Health and Human Services. Additional support was provided by U.S. Public Health Service Research grants AR052105 and AR048919. Dr. Lee is supported by National Center for Research Resources grant UL 1 RR024146.

Potential Financial Conflicts of Interest: Consultancies: J.A. Cauley (Novartis, Eli Lilly), D.C. Bauer (Merck, Roche Diagnostics), J. Wactawski-Wende (Johnson & Johnson), S.R. Cummings (Eli Lilly, Procter & Gamble, Amgen, GlaxoSmithKline, Zelos). Honoraria: J.A. Cauley (Eli Lilly), R.D. Jackson (Sanofi-Aventis, Procter & Gamble), J. Wactawski-Wende (Merck), S.R. Cummings (Eli Lilly). Stock ownership or options (other than mutual funds): M.S. LeBoff (Amgen, General Electric). Expert testimony: S.R. Cummings (Eli Lilly). Grants received: J.A. Cauley (Merck, Pfizer, Novartis), D.C. Bauer (Novartis, Amgen, Procter & Gamble, Merck), R.D. Jackson (MicroMRI, Procter & Gamble), M.S. LeBoff (Abbott), J. Ockene (National Heart, Lung, and Blood Institute), S.R. Cummings (Amgen, Eli Lilly).

Reproducible Research Statement: *Study protocol:* WHI study protocols are available at http://whiscience.org. *Statistical code and data set:* Not available.

Requests for Single Reprints: Jane A. Cauley, DrPH, University of Pittsburgh, Department of Epidemiology, 130 DeSoto Street, Crabtree A524, Pittsburgh, PA 15261; e-mail, jcauley@edc.pitt.edu.

Current author addresses and author contributions are available at www.annals.org.

References

1. Looker AC, Dawson-Hughes B, Calvo MS, Gunter EW, Sahyoun NR. Serum 25-hydroxyvitamin D status of adolescents and adults in two seasonal subpopulations from NHANES III. Bone. 2002;30:771-7. [PMID: 11996918] 2. Lips P. Vitamin D deficiency and secondary hyperparathyroidism in the elderly: consequences for bone loss and fractures and therapeutic implications. Endocr Rev. 2001;22:477-501. [PMID: 11493580]

3. Thomas MK, Lloyd-Jones DM, Thadhani RI, Shaw AC, Deraska DJ, Kitch BT, et al. Hypovitaminosis D in medical inpatients. N Engl J Med. 1998;338: 777-83. [PMID: 9504937]

4. LeBoff MS, Kohlmeier L, Hurwitz S, Franklin J, Wright J, Glowacki J. Occult vitamin D deficiency in postmenopausal US women with acute hip fracture. JAMA. 1999;281:1505-11. [PMID: 10227320]

5. Cranney A, Horsley T, O'Donnell S, Weiler H, Puil L, Ooi D, et al. Effectiveness and Safety of Vitamin D in Relation to Bone Health. Rockville, MD: Agency for Healthcare Research and Quality; 2007. Accessed at www.ahrq .gov/clinic/tp/vitadtp.htm on 17 June 2008.

6. Roddam AW, Neale R, Appleby P, Allen NE, Tipper S, Key TJ. Association between plasma 25-hydroxyvitamin D levels and fracture risk: the EPIC-Oxford study. Am J Epidemiol. 2007;166:1327-36. [PMID: 17716981]

7. Looker AC, Mussolino ME. Serum 25-hydroxyvitamin D and hip fracture risk in older U.S. white adults. J Bone Miner Res. 2008;23:143-50. [PMID: 17907920]

8. Dhesi JK, Jackson SH, Bearne LM, Moniz C, Hurley MV, Swift CG, et al.

Vitamin D supplementation improves neuromuscular function in older people who fall. Age Ageing. 2004;33:589-95. [PMID: 15501836]

9. Bischoff-Ferrari HA, Dawson-Hughes B, Willett WC, Staehelin HB, Bazemore MG, Zee RY, et al. Effect of vitamin D on falls: a meta-analysis. JAMA. 2004;291:1999-2006. [PMID: 15113819]

10. Semba RD, Garrett E, Johnson BA, Guralnik JM, Fried LP. Vitamin D deficiency among older women with and without disability. Am J Clin Nutr. 2000;72:1529-34. [PMID: 11101482]

11. Ensrud KE, Lui LY, Taylor BC, Ishani A, Shlipak MG, Stone KL, et al. Osteoporotic Fractures Research Group. Renal function and risk of hip and vertebral fractures in older women. Arch Intern Med. 2007;167:133-9. [PMID: 17242313]

12. Kestenbaum B, Belozeroff V. Mineral metabolism disturbances in patients with chronic kidney disease. Eur J Clin Invest. 2007;37:607-22. [PMID: 17635571]

13. Colin EM, Uitterlinden AG, Meurs JB, Bergink AP, van de Klift M, Fang Y, et al. Interaction between vitamin D receptor genotype and estrogen receptor alpha genotype influences vertebral fracture risk. J Clin Endocrinol Metab. 2003; 88:3777-84. [PMID: 12915669]

14. Heikkinen A, Parviainen MT, Tuppurainen MT, Niskanen L, Komulainen MH, Saarikoski S. Effects of postmenopausal hormone replacement therapy with and without vitamin D3 on circulating levels of 25-hydroxyvitamin D and 1, 25-dihydroxyvitamin D. Calcif Tissue Int. 1998;62:26-30. [PMID: 9405729]

15. Ooms ME, Roos JC, Bezemer PD, van der Vijgh WJ, Bouter LM, Lips P. Prevention of bone loss by vitamin D supplementation in elderly women: a randomized double-blind trial. J Clin Endocrinol Metab. 1995;80:1052-8. [PMID: 7714065]

16. Mezquita-Raya P, Muñoz-Torres M, Luna JD, Luna V, Lopez-Rodriguez F, Torres-Vela E, et al. Relation between vitamin D insufficiency, bone density, and bone metabolism in healthy postmenopausal women. J Bone Miner Res. 2001;16:1408-15. [PMID: 11499863]

17. Hays J, Hunt JR, Hubbell FA, Anderson GL, Limacher M, Allen C, et al. The Women's Health Initiative recruitment methods and results. Ann Epidemiol. 2003;13:S18-77. [PMID: 14575939]

18. Curb JD, McTiernan A, Heckbert SR, Kooperberg C, Stanford J, Nevitt M, et al. WHI Morbidity and Mortality Committee. Outcomes ascertainment and adjudication methods in the Women's Health Initiative. Ann Epidemiol. 2003;13:S122-8. [PMID: 14575944]

19. Patterson RE, Kristal AR, Tinker LF, Carter RA, Bolton MP, Agurs-Collins T. Measurement characteristics of the Women's Health Initiative food frequency questionnaire. Ann Epidemiol. 1999;9:178-87. [PMID: 10192650]

20. Manson JE, Greenland P, LaCroix AZ, Stefanick ML, Mouton CP, Oberman A, et al. Walking compared with vigorous exercise for the prevention of cardiovascular events in women. N Engl J Med. 2002;347:716-25. [PMID: 12213942]

21. Hays RD, Sherbourne CD, Mazel RM. The RAND 36-Item Health Survey 1.0. Health Econ. 1993;2:217-27. [PMID: 8275167]

22. Woods NF, LaCroix AZ, Gray SL, Aragaki A, Cochrane BB, Brunner RL, et al. Women's Health Initiative. Frailty: emergence and consequences in women aged 65 and older in the Women's Health Initiative Observational Study. J Am Geriatr Soc. 2005;53:1321-30. [PMID: 16078957]

23. Anderson GL, Manson J, Wallace R, Lund B, Hall D, Davis S, et al. Implementation of the Women's Health Initiative study design. Ann Epidemiol. 2003;13:S5-17. [PMID: 14575938]

24. Freeman RK, Bateman BG, Goebelsmann U, Arce JJ, James J. Clinical experience with the amniotic fluid lecithin-sphingomyelin ratio. II. The L-S ratio in "stressed pregnancies". Am J Obstet Gynecol. 1974;119:239-42. [PMID: 4823393]

25. Goebelsmann U, Arce JJ, Thorneycroft IH, Mishell DR Jr. Serum testosterone concentrations in women throughout the menstrual cycle and following HCG administration. Am J Obstet Gynecol. 1974;119:445-52. [PMID: 4842588]

26. Probst-Hensch NM, Ingles SA, Diep AT, Haile RW, Stanczyk FZ, Kolonel LN, et al. Aromatase and breast cancer susceptibility. Endocr Relat Cancer. 1999; 6:165-73. [PMID: 10731105]

27. Stanczyk FZ, Shoupe D, Nunez V, Macias-Gonzales P, Vijod MA, Lobo RA. A randomized comparison of nonoral estradiol delivery in postmenopausal women. Am J Obstet Gynecol. 1988;159:1540-6. [PMID: 3144919]

28. Rinaldi S, Geay A, Déchaud H, Biessy C, Zeleniuch-Jacquotte A, Akhmedkhanov A, et al. Validity of free testosterone and free estradiol determinations in

www.annals.org

serum samples from postmenopausal women by theoretical calculations. Cancer Epidemiol Biomarkers Prev. 2002;11:1065-71. [PMID: 12376508]

29. Södergård R, Bäckström T, Shanbhag V, Carstensen H. Calculation of free and bound fractions of testosterone and estradiol-17 beta to human plasma proteins at body temperature. J Steroid Biochem. 1982;16:801-10. [PMID: 7202083]

30. Vermeulen A, Verdonck L, Kaufman JM. A critical evaluation of simple methods for the estimation of free testosterone in serum. J Clin Endocrinol Metab. 1999;84:3666-72. [PMID: 10523012]

31. Hosmer DW, Lemeshow S. Applied Logistic Regression. 2nd ed. New York: J Wiley; 2000.

32. Gerdhem P, Ringsberg KA, Obrant KJ, Akesson K. Association between 25-hydroxy vitamin D levels, physical activity, muscle strength and fractures in the prospective population-based OPRA Study of Elderly Women. Osteoporos Int. 2005;16:1425-31. [PMID: 15744449]

33. Woo J, Swaminathan R, Pang CP, Mak YT, MacDonald D. A comparison of biochemical indices of bone turnover in elderly institutionalized and free-living subjects. Bone Miner. 1990;8:31-8. [PMID: 2306552]

34. Cummings SR, Browner WS, Bauer D, Stone K, Ensrud K, Jamal S, et al. Endogenous hormones and the risk of hip and vertebral fractures among older women. Study of Osteoporotic Fractures Research Group. N Engl J Med. 1998; 339:733-8. [PMID: 9731089]

35. Holick MF, Siris ES, Binkley N, Beard MK, Khan A, Katzer JT, et al. Prevalence of vitamin D inadequacy among postmenopausal North American women receiving osteoporosis therapy. J Clin Endocrinol Metab. 2005;90:3215-24. [PMID: 15797954]

36. Wortsman J, Matsuoka LY, Chen TC, Lu Z, Holick MF. Decreased bioavailability of vitamin D in obesity. Am J Clin Nutr. 2000;72:690-3. [PMID: 10966885]

37. Snijder MB, van Dam RM, Visser M, Deeg DJ, Dekker JM, Bouter LM, et al. Adiposity in relation to vitamin D status and parathyroid hormone levels:

a population-based study in older men and women. J Clin Endocrinol Metab. 2005;90:4119-23. [PMID: 15855256]

38. Macdonald HM, Mavroeidi A, Barr RJ, Black AJ, Fraser WD, Reid DM. Vitamin D status in postmenopausal women living at higher latitudes in the UK in relation to bone health, overweight, sunlight exposure and dietary vitamin D. Bone. 2008;42:996-1003. [PMID: 18329355]

39. Bischoff-Ferrari HA. The 25-hydroxyvitamin D threshold for better health. J Steroid Biochem Mol Biol. 2007;103:614-9. [PMID: 17227709]

40. Bischoff-Ferrari HA, Dietrich T, Orav EJ, Dawson-Hughes B. Positive association between 25-hydroxy vitamin D levels and bone mineral density: a population-based study of younger and older adults. Am J Med. 2004;116:634-9. [PMID: 15093761]

41. Bischoff-Ferrari HA, Willett WC, Wong JB, Giovannucci E, Dietrich T, Dawson-Hughes B. Fracture prevention with vitamin D supplementation: a meta-analysis of randomized controlled trials. JAMA. 2005;293:2257-64. [PMID: 15886381]

42. Grant AM, Avenell A, Campbell MK, McDonald AM, MacLennan GS, McPherson GC, et al. RECORD Trial Group. Oral vitamin D3 and calcium for secondary prevention of low-trauma fractures in elderly people (Randomised Evaluation of Calcium Or vitamin D, RECORD): a randomised placebo-controlled trial. Lancet. 2005;365:1621-8. [PMID: 15885294]

43. Jackson RD, LaCroix AZ, Gass M, Wallace RB, Robbins J, Lewis CE, et al. Women's Health Initiative Investigators. Calcium plus vitamin D supplementation and the risk of fractures. N Engl J Med. 2006;354:669-83. [PMID: 16481635]

44. Holick MF. Vitamin D deficiency. N Engl J Med. 2007;357:266-81. [PMID: 17634462]

45. Binkley N, Drezner MK, Hollis BW. Laboratory reporting of 25-hydroxyvitamin D results: potential for clinical misinterpretation [Letter]. Clin Chem. 2006;52:2124-5. [PMID: 17068180]

NEW PEER REVIEWERS

Sign up to become a peer reviewer for *Annals of Internal Medicine* by going to www.annals.org and selecting "Information for: Authors/ Reviewers." Then select "Reviewer Information" and register as a new reviewer. Note that *Annals* reviewers whose reviews are returned on time and are judged satisfactory by the Editors may receive up to 3 Category 1 CME credits per review (maximum, 15 credits in a calendar year).

Annals of Internal Medicine

Current Author Addresses: Dr. Cauley: University of Pittsburgh, Department of Epidemiology, 130 DeSoto Street, Crabtree A524, Pittsburgh, PA 15261.

Dr. LaCroix, Ms. L. Wu, and Ms. C. Wu: Fred Hutchinson Cancer Research Center, 1100 Fairview Avenue North, Seattle, WA 98109.

Dr. Horwitz: University of Pittsburgh Medical Center, Falk 580, 3601 Fifth Avenue, Pittsburgh, PA 15261.

Dr. Danielson: University of Pittsburgh, Department of Epidemiology, 130 DeSoto Street, Crabtree A543, Pittsburgh, PA 15261.

Dr. Bauer: University of California, San Francisco, 185 Berry Street, #5700, San Francisco, CA 94105.

Dr. Lee: University of California, Davis, 4150 V Street, Suite 6400, Sacramento, CA 75817.

Dr. Jackson: The Ohio State University, 485 McCampbell Hall, 1581 Dodd Drive, Columbus, OH 43210.

Dr. Robbins: Lawrence J. Ellison Ambulatory Care Center, 4860 Y Street, Sacramento, CA 95817.

Dr. Stanczyk: USC Keck School of Medicine, Women's & Children's Hospital, 1240 North Mission Road, Room 1M2, Los Angeles, CA 90033.

Dr. LeBoff: Brigham and Women's Hospital, Harvard Medical School, 221 Longwood Avenue, Boston, MA 02115.

Dr. Wactawski-Wende: State University of New York at Buffalo, 270 Farber Hall, Buffalo, NY 14214.

Dr. Sarto: University of Wisconsin-Madison, 700 Regent Street, Madison, WI 53715.

Dr. Ockene: University of Massachusetts Medical School, 55 Lake Avenue North, Worcester, MA 01655.

Dr. Cummings: San Francisco Coordinating Center, 185 Berry Street, Lobby 4, Suite 5700, San Francisco, CA 94107.

Author Contributions: Conception and design: J.A. Cauley, D.C. Bauer, R.D. Jackson, M.S. LeBoff, J. Wactawski-Wende, S.R. Cummings.

Analysis and interpretation of the data: J.A. Cauley, A.Z. LaCroix, M. Horwitz, L. Wu, D.C. Bauer, J.S. Lee, R.D. Jackson, C. Wu, M.S. LeBoff, G. Sarto.

Drafting of the article: J.A. Cauley, L. Wu, M. Horwitz, M.E. Danielson, M.S. LeBoff.

Critical revision of the article for important intellectual content: A.Z. LaCroix, L. Wu, M. Horwitz, M.E. Danielson, D.C. Bauer, J.S. Lee, R.D. Jackson, F.Z. Stanczyk, M.S. LeBoff, J. Wactawski-Wende, G. Sarto, S.R. Cummings.

Final approval of the article: J.A. Cauley, A.Z. LaCroix, L. Wu, M. Horwitz, M.E. Danielson, D.C. Bauer, R.D. Jackson, J.A. Robbins, F.Z. Stanczyk, J. Wactawski-Wende, J. Ockene, S.R. Cummings.

Provision of study materials or patients: J.A. Cauley, J.S. Lee, R.D. Jackson, J.A. Robbins, F.Z. Stanczyk, J. Wactawski-Wende, G. Sarto, J. Ockene.

Statistical expertise: L. Wu.

Obtaining of funding: J.A. Cauley, M.E. Danielson, J.S. Lee, R.D. Jackson, J.A. Robbins, J. Wactawski-Wende, J. Ockene, S.R. Cummings. Administrative, technical, or logistic support: J.A. Cauley, M.E. Danielson, J.S. Lee, R.D. Jackson, J. Wactawski-Wende.

Collection and assembly of data: J.A. Cauley, A.Z. LaCroix, J.S. Lee, R.D. Jackson, J.A. Robbins, J. Wactawski-Wende, G. Sarto, J. Ockene.

APPENDIX: THE WHI INVESTIGATORS

Program Office

National Heart, Lung, and Blood Institute, Bethesda, Maryland: Elizabeth Nabel, Jacques Rossouw, Shari Ludlam, Linda Pottern, Joan McGowan, Leslie Ford, and Nancy Geller.

www.annals.org

Clinical Coordinating Centers

Fred Hutchinson Cancer Research Center, Seattle, Washington: Ross Prentice, Garnet Anderson, Andrea LaCroix, Charles L. Kooperberg, Ruth E. Patterson, and Anne McTiernan.

Wake Forest University School of Medicine, Winston-Salem, North Carolina: Sally Shumaker.

Medical Research Labs, Highland Heights, Kentucky: Evan Stein.

University of California, San Francisco, San Francisco, California: Steven Cummings.

Clinical Centers

Albert Einstein College of Medicine, Bronx, New York: Sylvia Wassertheil-Smoller.

Baylor College of Medicine, Houston, Texas: Aleksandar Rajkovic.

Brigham and Women's Hospital, Harvard Medical School, Boston, Massachusetts: JoAnn Manson.

Brown University, Providence, Rhode Island: Annlouise R. Assaf.

Emory University, Atlanta, Georgia: Lawrence Phillips.

Fred Hutchinson Cancer Research Center, Seattle, Washington: Shirley Beresford.

George Washington University Medical Center, Washington, DC: Judith Hsia.

Los Angeles Biomedical Research Institute at Harbor-UCLA Medical Center, Torrance, California: Rowan Chlebowski.

Kaiser Permanente Center for Health Research, Portland, Oregon: Evelyn Whitlock.

Kaiser Permanente Division of Research, Oakland, California: Bette Caan.

Medical College of Wisconsin, Milwaukee, Wisconsin: Jane Morley Kotchen.

MedStar Research Institute/Howard University, Washington, DC: Barbara V. Howard.

Northwestern University, Chicago/Evanston, Illinois: Linda Van Horn.

The Ohio State University, Columbus, Ohio: Rebecca Jackson.

Rush Medical Center, Chicago, Illinois: Henry Black.

Stanford Prevention Research Center, Stanford, California: Marcia L. Stefanick.

State University of New York at Stony Brook, Stony Brook, New York: Dorothy Lane.

University of Alabama at Birmingham, Birmingham, Alabama: Cora E. Lewis.

University of Arizona, Tucson/Phoenix, Arizona: Tamsen Bassford.

University at Buffalo, Buffalo, New York: Jean Wactawski-Wende.

University of California, Davis, Sacramento, California: John Robbins.

University of California, Irvine, Irvine, California: F. Allan Hubbell.

University of California, Los Angeles, Los Angeles, California: Lauren Nathan.

University of California, San Diego, LaJolla/Chula Vista, California: Robert D. Langer.

University of Cincinnati, Cincinnati, Ohio: Margery Gass. University of Florida, Gainesville/Jacksonville, Florida: Marian Limacher.

University of Hawaii, Honolulu, Hawaii: David Curb. University of Iowa, Iowa City/Davenport, Iowa: Robert Wallace.

University of Massachusetts/Fallon Clinic, Worcester, Massachusetts: Judith Ockene.

University of Medicine and Dentistry of New Jersey, Newark, New Jersey: Norman Lasser.

University of Miami, Miami, Florida: Mary Jo O'Sullivan.

University of Minnesota, Minneapolis, Minnesota: Karen Margolis.

University of Nevada, Reno, Nevada: Robert Brunner.

University of North Carolina, Chapel Hill, North Carolina: Gerardo Heiss.

University of Pittsburgh, Pittsburgh, Pennsylvania: Lewis Kuller.

University of Tennessee, Memphis, Tennessee: Karen C. Johnson.

University of Texas Health Science Center, San Antonio, Texas: Robert Brzyski.

University of Wisconsin, Madison, Wisconsin: Gloria E. Sarto. Wake Forest University School of Medicine, Winston-Salem, North Carolina: Denise Bonds.

Wayne State University School of Medicine/Hutzel Hospital, Detroit, Michigan: Susan Hendrix.